Author’s response to reviews

Title: Fecal microbiota transplantation for severe clostridium difficile infection after left ventricular assist device implantation: a case control study and concise review on the local and regional therapies

Authors:

Zeina Berro (drzeinaberro@gmail.com)
Righab Hamdan (mldrighabh@hotmail.com)
Israa Dandache (ihd01@mail.aub.edu)
Mohamad Saab (israadandache@gmail.com)
Hussein Karnib (h.karnib@gmail.com)
Mahmoud Younes (myouness55@yahoo.com)

Version: 2 Date: 27 Apr 2016

Author’s response to reviews:

Dear Editor,

Please find our revised manuscript. We would like to thank the Managing Editor and the Reviewers for their helpful comments and suggestions to improve our manuscript.

The manuscript has been thoroughly revised to include all the needed changes.